Daily status, June 9

SOAS ISFS Status, June 9, 2013

Summary: 32m Licor out; 20m sonic looks okay after 01:00 last night. Soil.a okay except for Tsoil.4.4cm.a

Sonics: ok; 20m sonic now looks okay

TRH, P, p: Ifan.8m=0; restarted at 9:23 CDT

Wind speed & direction: ok

Soil, Rainr: soil.a mote as been working since Chris gave it love yesterday, however Tsoil.4.4cm now bad.
Vmote.(rad,a,b) ~12.5, 19.9, 11.4 V.

Radiation, Wetness: ok

h20, co2: 32m Licor not reporting. kh2oV.(3m, 13.9m) ~ 0.37, 0.33 V

System (GPS, Vdsm): ok
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